Author’s response to reviews

Title: Factors affecting the cultural competence of visiting nurses for rural multicultural family support in South Korea

Authors:
Minhyun Suk (chnursing@cha.ac.kr)
Won-Oak Oh (wooh@korea.ac.kr)
YeoJin Im (imyj@khu.ac.kr)

Version: 3 Date: 17 Nov 2017

Author’s response to reviews:
Comments from Editor:

Dear Editor
Thank you for your comments. We It was very helpful to improve the quality of our paper.

(1) pages 11 and 12, in some cases you use the plus/minus symbol before SD (and not in other cases). I suggest to omit the plus/minus symbol and use SD=→
 ==> According to Editor’s, we changed from ‘±SD’ to ‘SD =’ in page 11 and 12 and left them in blue. Thank you so much for your comments as always.